Author’s response to reviews

Title: Laryngeal Mask versus Facemask in the Respiratory Management during Catheter Ablation

Authors:

Takashi Koyama (takashixkoyama@icloud.com)
Masanori Kobayashi (kobayashima@chushin-miniren.gr.jp)
Tomohide Ichikawa (ichikawat@chushin-miniren.gr.jp)
Yasushi Wakabayashi (wakabayashiy@chushin-miniren.gr.jp)
Daiki Toma (TomaD@chushin-miniren.gr.jp)
Hidetoshi Abe (abeh@chushin-miniren.gr.jp)

Version: 2 Date: 23 Dec 2019

Author’s response to reviews:

<;Reply to the Reviewer>
Revision “BANE-D-19-00779 R1”

The authors are grateful for the helpful comments and constructive suggestions of this reviewer. We have revised the manuscript according to the reviewer’s suggestions. The following comments are in response to each of the reviewer’s points:

1. In the section 'Ethics Approval and Consent to Participate', please state whether the informed consent obtained was written or verbal. If verbal, please state the reason and whether the ethics committee approved this procedure.
   → Thank you for helpful comments. We added the section 'Ethics Approval and Consent to Participate' and describe that written informed consent was performed in this study.

2. Please add a “Declarations” heading after the Abbreviations section.
   → Thank you for helpful comments. We add a “Declarations” heading after the Abbreviations section according to the suggestions.

3. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions.
   → Thank you for your suggestions. We uploaded final clean version at this stage.
Please ensure all figures, tables and additional files are cited in the text, as they may be removed by the production department. Please delete any additional files not intended for publication.

We confirmed main document, figures and tables at this stage.